Author's response to reviews

Title: Biomechanical comparison of anterior lumbar interbody fusion: stand-alone interbody cage versus interbody cage with pedicle screw fixation - a finite element analysis

Authors:

Kyung-Chul Choi (chul5104@hanmail.net)
Kyeong-Sik Ryu (chul5104@hanmail.net)
Sang-Ho Lee (chul5104@hanmail.net)
Yeong Hyeon Kim (chul5104@hanmail.net)
Sung Jae Lee (chul5104@hanmail.net)
Chun-Kun Park (ckpmd@catholic.ac.kr)

Version: 4 Date: 31 May 2013

Author's response to reviews:

I revised a manuscript as follows:

Competing interests
The authors declare that they have no competing interests.

Authors' contributions
Ryu KS, Lee SH, Park CK provided advice on the study design. Choi KC carried out the whole studies, participated in the sequence alignment and wrote the article. And Choi KC was responsible for the data acquisition/analysis, interpretation. Lee SH provided advice on the data analysis. Ryu KS, Lee SH, Park CK contributed to the content of the article. Kim YH, Lee SJ involved in surgical treatment.
All authors read and approved the final manuscript.

Acknowledgements
This study was supported by a grant from the Wooridul Spine Foundation.